Author's response to reviews

Title: The effect of Melatonin on bacterial translocation following ischemia/reperfusion injury in a rat model of superior mesenteric artery occlusion

Authors:

Murat Ozban (muratozban@yahoo.com)
Cagatay Aydin (cagatayaydin@yahoo.com)
Nural Cevahir (cevahir@pau.edu.tr)
Cigdem Yenisey (mozban@pau.edu.tr)
Onur Birsen (birsenonur@yahoo.com)
Gulistan Gumrukcu (caydin@pau.edu.tr)
Berrin Aydin (berrinaydin@hotmail.com)
Ibrahim Berber (iberber@hotmail.com)

Version: 6 Date: 16 January 2015

Author's response to reviews: see over
Editorial comments:

- Thank you for including a Title Page at the beginning of your manuscript. However, we would ask you to please include the email addresses of all authors in this section.

1. We included all email addresses of all authors in this section

- Please remove your figures from your manuscript document. The figures that you have uploaded into our submission system are automatically included at the end of your manuscript when the PDF version generate. Figures legends however should be included in your manuscript, and not uploaded.

2. We removed figures from our manuscript. Figure legends are included at the end of the manuscript. Table legends are included at the end of the manuscript.

- Similarly to the above point, please remove your tables from the figures section of our submission system. Tables should only be included in your manuscript document.

3. We removed tables from the submission system.

- Thank you for including an Acknowledgements section in your manuscript. However, this appears to include information on the contributions of authors to your manuscript, rather than acknowledging any assistance that was received from persons other than the authors. If you have no-one else to acknowledge, then we would suggest simply removing this section from your manuscript.

4. We removed acknowledgments section in our manuscript.

Thank you very much for your interesting to our manuscript.

Sincerely.

Murat Ozban